Author’s response to reviews

Title: The analysis of relationship between selected sociodemographic factors and disorders of speech organs in Parkinson's patients

Authors:
Wioletta Pawlukowska (wsna@o2.pl)
Karolina Skonieczna-Żydecka (karzyd@sci.pum.edu.pl)
Iwona Rotter (iwrot@wp.pl)
Krystyna Honczarenko (khonczar@sci.pum.edu.pl)
Przemysław Nowacki (przemyslaw.nowacki@pum.edu.pl)

Version: 1 Date: 06 Sep 2017

Author’s response to reviews:

Cover letter a response

Editor Comments

1. We have improved the criteria for inclusion and exclusion criteria - as recommended by the reviewer

2. Our study did not show the relationship between the disease severity, the disease duration and the dose of L-dopa and articulation in patients with PD;

3. We have supplemented the FDA test information - as recommended by the reviewer;

Dear Reviewer 1 Elmar Nöth

Thank you very much for your valuable comments
Abstract:

1. We corrected the errors and supplemented the missing literature - as recommended by the reviewer;

Material and method:

1. We have improved the criteria for inclusion and exclusion criteria - as recommended by the reviewer

2. We corrected the errors - as recommended by the reviewer

Statistical analysis:

1. We corrected the errors - as recommended by the reviewer

Results:

1. We corrected the errors - as recommended by the reviewer

2. As far as statistical analysis is concerned. The number of patients in each rating of particular speech determinants in study group has been added to the table.

3. The analysis was done by means of Kruskall-Wallis test (One-way ANOVA on ranks), in which “Mean” denotes Mean Rank and H is test statistic value
Discussion

1. We corrected the errors as recommended by the reviewer

2. We completed the last part of the discussion as suggested by the reviewer